Author’s response to reviews

Title: Respiratory-gated (4D) contrast-enhanced FDG PET-CT for radiotherapy planning of lower oesophageal carcinoma: feasibility and impact on planning target volume

Authors:

Andrew Scarsbrook (a.scarsbrook@nhs.net)
Gillian Ward (gillian.ward11@nhs.net)
Patrick Murray (patrick.murray@nhs.net)
Rebecca Goody (r.goody@nhs.net)
Karen Marshall (karen.marshall21@nhs.net)
Garry McDermott (garry.mcdermott@nhs.net)
Robin Prestwich (robin.prestwich@nhs.net)
Ganesh Radhakrishna (g.radhakrishna@nhs.net)

Version: 3 Date: 31 Aug 2017

Author’s response to reviews:

Editor Comments:

1. Please remove the response to reviewers from the file inventory, as it is no longer needed at this stage of the editorial process.

This has been removed as requested

2. Please move the list of abbreviations to the end of the manuscript after the Conclusions section.

These have been moved as requested

3. Please move the statement regarding informed human subject request from the ‘Consent for publication’ section to the ‘Ethics approval and consent to participate’ section.
The Consent for Publication subsection concerns the publication of personally identifiable data. Since your study does not publish such data, please remove your current statement, and state “Not Applicable”.

This has been amended as requested

4. In the Funding section, please indicate the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The funding body reviewed a grant application which included the proposed study outline, but have had no role in study design, data collection, analysis, interpretation or in writing the manuscript. This has been clarified in the funding section as requested.